

Review



Oral Antiviral Treatment for COVID-19: A Comprehensive Review on Nirmatrelvir/Ritonavir

Karolina Akinosoglou *⁰, Georgios Schinas ⁰ and Charalambos Gogos

Department of Internal Medicine, Medical School, University of Patras, 26504 Rio, Greece * Correspondence: akin@upatras.gr; Tel.: +30-697-776-2897

* Correspondence: akin@upatras.gr; Iel.: +30-697-776-289

Abstract: Despite the rapid development of efficient and safe vaccines against COVID-19, the need to confine the pandemic and treat infected individuals on an outpatient basis has led to the approval of oral antiviral agents. Taking into account the viral kinetic pattern of SARS-CoV-2, it is of high importance to intervene at the early stages of the disease. A protease inhibitor called nirmatrelvir coupled with ritonavir (NMV/r), which acts as a CYP3A inhibitor, delivered as an oral formulation, has shown much promise in preventing disease progression in high-risk patients with no need for supplemental oxygen administration. Real-world data seem to confirm the drug combination's efficacy and safety against all viral variants of concern in adult populations. Although, not fully clarified, viral rebound and recurrence of COVID-19 symptoms have been described following treatment; however, more data on potential resistance issues concerning the Mpro gene, which acts as the drug's therapeutic target, are needed. NMV/r has been a gamechanger in the fight against the pandemic by preventing hospitalizations and halting disease severity; therefore, more research on future development and greater awareness on its use are warranted.

Keywords: COVID-19; nirmatrelvir/ritonavir; Paxlovid; SARS-CoV-2; nirmatrelvir; ritonavir; treatment

1. Introduction

Severe acute respiratory syndrome coronavirus 2, most commonly known as SARS-CoV-2, is an RNA virus first detected in Wuhan, China, in December 2019 that has posed a significant burden on health systems worldwide with its spread, which led to the coronavirus disease 2019 (COVID-19) pandemic. In the following two years, the coronavirus disease resulted in significant mortality and morbidity, especially in people with chronic conditions and multiple comorbidities. Clinical manifestations regarding COVID-19 are mainly driven by the biphasic nature of the disease, including an initial viral replication and toxicity stage, followed by a second phase of inflammatory response [1]. The latter is responsible for the complicated disease, as reflected by severe pneumonia, respiratory failure, the need for hospitalization, mechanical ventilation, and death.

Following SARS-CoV-2 infection, approximately 81% of patients will develop a mild disease that, in the majority of cases, will end in symptom resolution and complete recovery, while in 33%, at least one persistent symptom will occur, constituting long COVID syndrome [2,3]. In 14% of infected patients, however, especially in those with predisposing factors, such as age > 60 years, increased body mass index (BMI), chronic comorbidities, and immunosuppression [4,5], clinical deterioration is common, leading to intensive care unit admission and death, resulting in an overall COVID-19 case-fatality rate of 2.3% [6] and a hospital case-fatality rate of 13.6% [7].

As of April 2022, 65% of the world population has received at least one dose of a COVID-19 vaccine. Despite this breakthrough, COVID-19 will constitute an enduring public health issue [8]. Vaccine hesitancy [9], breakthrough infections [10], waning immunity [11], new variants [12], and persistent transmission despite vaccination [13] represent only some of the reasons that vaccines alone cannot completely control the pandemic [8].



Citation: Akinosoglou, K.; Schinas, G.; Gogos, C. Oral Antiviral Treatment for COVID-19: A Comprehensive Review on Nirmatrelvir/Ritonavir. *Viruses* **2022**, 14, 2540. https://doi.org/10.3390/ v14112540

Academic Editor: Yinzhong Shen

Received: 23 October 2022 Accepted: 15 November 2022 Published: 17 November 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). Improvements to other aspects of management can supplement vaccines' actions and efficacy. These may include new approaches in preventing infection by containing transmission from asymptomatic individuals [1] or protecting exposed individuals from infection [14], but also in rapid diagnosis and outpatient oral alternatives [15]. In early 2021, EMA approved the first oral antiviral against COVID-19, i.e., nirmatrelvir/ritonavir (NMV/r) [16]. We aim to review current data pertaining to the present experience and future directions regarding the use of nirmatrelvir/ritonavir in clinical practice

2. Methods

Broad searches of PubMed, Scopus, and Embase between 1 January 2020 and 10 November 2022 were conducted using the following keywords: 'nirmatrelvir/ritonavir', 'nirmatrelvir', and 'Paxlovid'. Relevant publications were identified based on the titles and abstracts. No restriction on type of paper was set. Only English language papers were included in this review and the main focus was put on clinical data. Two reviewers independently screened all title/abstracts and hand-searched references of retrieved articles. Disagreements were discussed and duplicates were removed.

3. Opportunities for Early Intervention: Nirmatrelvir/Ritonavir

A stereotypical viral kinetic pattern characterizes SARS-CoV-2 infection with a high peak viral load during the first several days of infection, a subsequent short rapid decay period followed by a slower clearance phase of variable duration, and, finally, a rapid elimination phase [17]. In the past, antiviral medications have demonstrated efficacy in treating various infectious diseases in the early stages of disease, reducing morbidity, mortality, and the likelihood of onward transmission [18]. Concerning COVID-19, modeling studies predicting the effects of antiviral treatments suggest better outcomes with an early approach. Assuming the viral shedding duration is a relevant surrogate of COVID-19 severity, an early test-and-treat approach is highly promising for limiting the likelihood of severe disease [17]. In order to reduce peak viral load (measured by RT-PCR and observed at day 5 post onset of symptoms) by more than two logs, drug efficacy needs to be >90% if treatment is administered after symptom onset; an efficacy of 60% could be sufficient only if treatment is initiated before symptom onset [19]. A systematic review and meta-analysis showed that the mean incubation period of COVID-19 was 5.74 days [20], further strengthening the argument for early treatment.

Therapeutic targets in the viral life cycle are based on four pillars [21]: (A) binding, i.e., preventing SARS-CoV-2 binding to ACE2 receptor (mAbs, convalescent plasma or plasmaderived therapies, and vaccines fall within this category); (B) entry/exit, i.e., preventing viral/host interaction and endosome maturation (for example, ACE2 inhibitors and human protease inhibitors); (C) proteolysis, i.e., prevention of protein formation involved in viral replication with protease inhibitors of 3CLpro or papain-like protease (PLpro), which will be our main topics of discussion herein; and (D) RNA replication, i.e., prevention of viral genome replication through RNA polymerase inhibitors.

Protease inhibitors (PIs) are precedented to treat hepatitis C (HCV) and human immunodeficiency viruses (HIV) (Figure 1) [22] by blocking the essential processing of viral proteins for replication.

The precedent suggests that protease inhibitors work clinically at concentrations above those shown to eradicate 90% of the virus in *in vitro* assays [23]. A correlation between viral load reduction and *in vitro* drug concentration for HCV and HIV has been noted, and viral load measurement may be used as a clinical measure of efficacy [24–26]. In addition, proteases are not subject to immune escape, unlike the spike protein of SARS-CoV2, and therefore protease inhibitors are predicted to be effective against variants [27].



Figure 1. Protease inhibitors (PIs) evolution.

PF-07321332, also known as nirmatrelvir, has been designed to inhibit Mpro, one of the most attractive drug targets among coronaviruses, since it is essential for viral replication and its substrate-recognition pocket is highly conserved [23,24,28]. PF-07321332 binds to the active site of Mpro and forms a reversible covalent bond with the catalytic cysteine residue, Cys145. Inhibition of Mpro renders the protein incapable of processing polyprotein precursors, thereby preventing viral replication [24,25,28]. The 3CL recognition sequence (Leu-Gln↓Ser-Ala-Gly) is not associated with any known human proteases, hence theoretically avoiding interaction with human cells. PF-07321332 demonstrated antiviral activity against all variants of concern, including Omicron and its variants [29].

Ritonavir (RTV) is an oral antiretroviral medication originally developed to treat HIV that constitutes a strong CYP3A inhibitor [30]. RTV thus enhances the bioavailability of nirmatrelvir by slowing its metabolism by CYP34A and is not expected per se to mediate any antiviral activity against SARS-CoV-2 [24]. Most common side effects are observed at high doses (600 mg BID); hence, only a low dose is used in combination with other PIs as a pharmacokinetic (PK) enhancer [30].

4. Administration

Patients receive the NMV/r (300 mg/100 mg) treatment course orally every 12 h for 5 days [31]. The regimen should be administered as soon as possible after the diagnosis of COVID-19 and within 5 days of symptom onset. The full 5-day treatment course should be completed, even if the patient requires hospitalization due to severe or critical COVID-19 after starting treatment with NMV/r.

Two pink 150 mg tablets of PF-07321332 and one white 100 mg tablet of ritonavir should be taken concurrently, with or without food. The tablets must be swallowed whole and not chewed, broken, or crushed, as no data are currently available. If the patient misses a dose of NMV/r within 8 h of the time it is usually taken, the patient should take it as soon as possible and resume the normal dosing schedule. If the patient misses a dose by more than 8 h, the patient should not take the missed dose and instead take the next dose at the regularly scheduled time. Double dosing to make up for a missed dose is not recommended.

No dose adjustment is needed in patients with mild renal impairment (eGFR ≥ 60 to <90 mL/min). However, the regimen should be avoided in patients with severe renal impairment (eGFR < 30 mL/min, including patients with end-stage renal disease (ESRD) under hemodialysis). In patients with moderate renal impairment (eGFR ≥ 30 to <60 mL/min), the dose of NMV/r should be reduced to 150 mg/100 mg, twice a day for 5 days, to avoid overexposure (even though no clinical data are available for this dose adjustment). Mild or moderate hepatic impairment does not require dose adjustment, but the regimen should not be used in patients with severe hepatic impairment.

5. A Word on Interactions

NMV/r is an inhibitor of CYP3A; hence, medicinal products that are primarily cleared by CYP3A are prone to significant increases in exposure when co-administered. In such cases, products highly metabolized by CYP3A and for which elevated plasma levels are related to severe and/or life-threatening events are contraindicated.

In addition, ritonavir has a high affinity for several CYP isoforms and may inhibit oxidation with the following ranked order: CYP3A4 > CYP2D6. As a result, it may induce glucuronidation and oxidation by CYP1A2, CYP2C8, CYP2C9, and CYP2C19, leading to enhanced biotransformation of some medicinal products associated with these pathways and resulting in decreased systemic exposure to some agents. In turn, the therapeutic effect of co-administrated agents could be decreased or shortened.

Thus, coadministration of other CYP3A isoform substrates that could lead to potentially significant interaction should be weighed against the risks. As a conservative measure, the drug–drug interactions pertaining to ritonavir used in chronic HIV infection should apply for NMV/r. Online tools to check for interactions remain available, and regimen adjustment is recommended upon clinical judgment and on a case-by-case basis. (Figure 2) [32].



Figure 2. Algorithm on potential drug–drug interactions of nirmatrelvir/ritonavir.

6. Guidelines

Currently, WHO and NIH strongly recommend NMV/r for those with a high risk of hospital admission, mostly in the context of outpatient treatment [4,33]. Other options include molnupiravir, remdesivir, and sotrovimab [32]. Therapeutic choices and characteristics are shown in Table 1. Nirmatrelvir/ritonavir represents a superior choice over other treatment options for those with non-severe illness at the highest risk. It may prevent more hospitalizations than the alternatives, has fewer harms than molnupiravir, and is easier to administer than IV options (remdesivir, monoclonal antibodies) [34].

	Absolute Risk Reduction	Relative Risk Reduction	Drug Interactions	Administration	Pregnancy	Adverse Events
Nirmatrelvir/ ritonavir	6.3→0.8%	88%	Anticipated drug interactions of concern	Oral	Safe in pregnancy	Minor
Remdesivir	5.3→0.7%	87%	Few/no anticipated drug interactions	Intravenous	Safe in pregnancy	Minor
Molnupiravir	9.7→6.8%	30%	Few/no anticipated drug interactions	Oral	Not recommended in pregnancy	Mutagenicity
Sotrovimab	7→1%	85%	Few/no anticipated drug interactions	Intravenous	Safe in pregnancy	Minor

Table 1. Comparison of treatment options for high-risk non-hospitalized patients.

7. Clinical Trial Data

NMV/r was the first per os antiviral COVID-19 medication authorized in the EU, following clinical trial results. Evaluation of Protease Inhibition for COVID-19 in High-Risk Patients (EPIC-HR) was a study of oral PF-07321332/ritonavir compared with placebo in non-hospitalized high-risk adults with COVID-19 [27]. Participants were randomized 1:1 to receive either NMV/r or placebo every 12 h for five days, while randomization was stratified by geographic region and receipt or expected receipt of mAb treatment for COVID-19. EPIC-HR assessed a composite primary endpoint of COVID-19-related hospitalization or death from any cause through day 28. Between 16 July and 9 December 2021, 2246 patients were enrolled from 343 centers worldwide. Most patients (93.8%) had not received or were not expected to receive mAb treatment for COVID-19, while 66.3% received the first dose of the trial drug or placebo within 3 days after symptom onset. A total of 61.0% of participants had \geq 2 characteristics/comorbidities associated with risk of progression, including BMI ≥ 25 kg/m² (80.5%), cigarette smoking (39.0%), hypertension (32.9%), >65 years of age (12.8%), and diabetes mellitus (12.2%) [27]. NMV/r resulted in significantly fewer COVID-19-related hospitalization or death events by day 28 versus placebo (p < 0.001). The relative risk reduction (RRR) was 89.1% and 88.9% at the interim and final analysis timepoints, respectively. NMV/r resulted in an approximately 10-fold reduction in viral load over time versus placebo when treatment was initiated within 3 days after symptom onset. The incidence of AEs that emerged during or after the treatment period (TEAEs) was similar among recipients of NMV/r (22.6%) and the placebo (23.9%). Most TEAEs in both treatment groups were mild to moderate (Grades 1–2) in severity. No patients in the NMV/r group experienced an AE resulting in death (Grade 5), while there were 13 deaths among placebo recipients.

8. Real-World Evidence

Real-world data on effectiveness have confirmed EPIC-HR primary results showing significant risk reduction in hospitalization and death [35–48] (Table 2).

Reference	Country of Origin	Population Characteristics	Outcomes
Arbel et al., 2022; Oral Nirmatrelvir and Severe COVID-19 Outcomes During the Omicron Surge [35] https://doi.org/10.21203/rs.3.rs-1705061/v1	Israel	 Omicron B1 variant Outpatients (109,213 eligible, 3939 treated) 64% were ≥65 78% prior immunity (vaccinated or infection) 	Primary endpoint: COVID-19 hospitalization Secondary endpoint: Death 68% RRR for COVID-19 hospitalization > 65 years with prior immunity and 85% without prior immunity; 79% RRR for death in people aged >65 years No impact on younger patients (confounders noted)

Table 2. Real-world data on the effectiveness of PaxlovidTM.

 Table 2. Cont.

Reference	Country of Origin	Population Characteristics	Outcomes
Najjar-Debbiny et al., 2022; Effectiveness of Paxlovid in Reducing Severe COVID-19 and Mortality in High-Risk Patients [36] https://doi.org/10.1093/cid/ciac443	Israel	 Omicron B1 variant Outpatients (180,35 eligible, 4737 treated) 79% were ≥60 75% adequately vaccinated 	Composite Endpoint: Severe COVID-19 or COVID-19 specific mortality 46% RRR for Pax, 80% RRR for vaccine Effectiveness similar in vaccinated and unvaccinated
Wong et al., 2022; Real-world effectiveness of molnupiravir and nirmatrelvir; ritonavir among COVID-19 inpatients during Hong Kong's Omicron BA.2 wave: an observational study [37] https://doi.org/10.1016/S1473-3099(22)00507-2	Hong Kong	 Omicron B2 variant Hospitalized patients not initially O₂-dependent (1856 molnupiravir recipients, 890 Pax recipients and matched controls) 82% were ≥65 10% fully vaccinated, no prior COVID infections 	Composite Endpoint: death, IMV, or ICU admission 43% reduction in composite endpoint for Pax Shorter length of hospital stays in survivors 40% reduction in composite endpoint for molnupiravir
Wong et al., 2022; Real-world effectiveness of molnupiravir and nirmatrelvir; ritonavir against mortality, hospitalization, and in-hospital outcomes among community-dwelling, ambulatory COVID-19 patients during the BA.2.2 wave in Hong Kong: an observational study [38] https://doi.org/10.1108/2022.05.26.22275631	Hong Kong	 Omicron B2 variant Non-hospitalized patients (1072,004 eligible, 5257 received molnupiravir and 5663 Pax) 68.5% were ≥65 35% fully vaccinated 	Endpoints: Death, COVID-19-related hospitalization, and composite of in-patient death, IMV, or ICU admission 31% reduction in hospitalization; 75% reduction in death; protection of Pax > molnupiravir
Yip et al., 2022; Impact of the use of oral antiviral agents on the risk of hospitalization in community COVID-19 patients [39] https://doi.org/10.1093/cid/ciac687	Hong Kong	 Omicron B2 variant Non-hospitalized patients (83,154 controls, 5808 molnupiravir, 4921 Pax recipients) Average age was 71 years old 43% fully vaccinated 	Primary endpoint: Hospitalization Secondary endpoint: Death, IMV, or ICU admission 21% reduction in hospitalization with Pax, no impact with molnupiravir No difference for secondary endpoint
Ganatra S et al., 2022; Oral Nirmatrelvir and Ritonavir in Non-hospitalized Vaccinated Patients with COVID-19 [41] https://doi.org/10.1093/cid/ciac673	USA	 Vaccinated Non-hospitalized (1130 patients and equally matched controls) Mean age 57.6 years, 63% female 	Primary outcome: Composite ER visits, hospitalization, or death Secondary endpoints: Individual components and symptoms 45% RRR in primary composite outcome Ten deaths noted, all in the non-Pax cohort
Dryden-Petersonet, S et al., 2022; Nirmatrelvir plus ritonavir for early COVID-19 and hospitalization in a large US health system [40] https://doi.org/10.1108/2022.06.14.22276393	USA	 Omicron B2 variant (lineages BA.1.1, BA.2, and BA.2.12.1) Non-hospitalized patients (24,286 controls, 6036 were prescribed Pax) Individuals aged 50 years and older 87.2% vaccinated 	Primary endpoint: Hospitalization D14 Hospitalization in 40 (0.66%) with Pax and 232 (0.96%) control (adjusted RR: 0.55 (95% CI: 0.38 to 0.80, p = 0.002)) 39 deaths control vs. no deaths Pax Low baseline rate with greatest impact in unvaccinated and obese patients
Malden DE et al., 2022; Hospitalization and Emergency Department Encounters for COVID-19 After Paxlovid Treatment—California, December 2021–May 2022 MMWR (cdc.gov) [42]	USA	 Outpatient population (n = 5287) 47% were >65 y 73% vaccinated with 3 doses 56.7% Pax recipients with >1 medical condition 	Endpoints: Hospitalization or ER COVID-19-related illness during the 5–15 days after Pax < 1% of all patients 6 patients hospitalized and 39 ER were either unvaccinated or vaccinated with 1 dose (17.8%) vs. all treated patients (9.7%), 2 deaths (multiple comorbidities and age)

Table 2. Cont.

Reference	Country of Origin	Population Characteristics	Outcomes
Allen S et al., 2022; Game Changer: Paxlovid Reduces Hospitalization and Saves Lives (epicresearch.org) [43]	USA	 Cosmos Data December 2021–August 2022 N(Pax) = 29,334 N(control) = 1,963,129 	Endpoints: Hospital admission, death, or discharge Hospitalization: Pax 1.86% vs. control 9.6% Death: Pax 0.12% vs. control 1.23% Patients prescribed Pax ~5× less likely to be hospitalized and 10× less likely to die after COVID-19 Patients > 45 years much more likely to be prescribed Nirmatrelvir/r
Lewnard et al., 2022; Effectiveness of nirmatrelvir-ritonavir against hospital admission: a matched cohort study in a large US healthcare system [44] https://doi.org/10.1101/2022.10.02.22280623	USA	 Non-hospitalized patients (4329 received Pax, 20,980 served as control) 93.3% vaccinated with 2 doses 78.1% vaccinated with 3 doses 94.2% symptomatic 	Primary endpoints: Effectiveness against hospital admission and acute respiratory infection (ARI)-associated hospital admission over 15 and 30 days Pax dispensed within 0–5 days of symptom onset; 88.1% (95% CI: 49.0-97.5%) over 15 days and 71.9% (25.3–90.0%) over 30 days effectiveness in preventing any admission 88.3% (12.9–98.8%) over 15 days and 87.3% (18.3–98.5%) over 30 days effectiveness in preventing ARI-associated hospital admissions Pax initiation at any point of disease; 86.6% (54.9–96.3%) and 78.0% (46.2–91.4%) effectiveness in preventing all hospital admissions over 15 and 30 days, respectively 93.7% (52.5–99.4%) and 92.8% (53.9–99.1%) in preventing ARI-associated hospital admissions over 15 and 30 days, respectively Similar effectiveness regardless of total vaccine doses (≥2 doses)
Gentile et al., 2022; Nirmatrelvir/Ritonavir and Molnupiravir in the Treatment of Mild/Moderate COVID-19: Results of a Real-Life Study [46] https://doi.org/10.3390/vaccines10101731	Italy	 Omicron wave 257 outpatients (56.8% received molnupiravir, 43.2% received Pax) 96.1% vaccinated 	Endpoints: 1.6% overall hospitalization rate over 14 days 2.1% hospitalization in molnupiravir group vs. 0.9% in Pax 12.1% reported at least one ADR. No statistically significant difference in ADRs reported between groups (8.9% molnupiravir vs. 16.2% Pax, p = 0.075)
Qian et al., 2022; Outcomes with and without outpatient SARS-CoV-2 treatment for patients with COVID-19 and systemic autoimmune rheumatic diseases: A retrospective cohort study [48] https://doi.org/10.1101/2022.10.27.22281629	USA	 Outpatients with systemic autoimmune rheumatic disease (N = 704,426 patients received treatment, 307 received Pax, 105 monoclonal Abs, 5 molnupiravir, 3 remdesivir, 6 combined treatment) 	Endpoint: Outpatient treatment reduced the odds of severe COVID-19 by 88% 8% of those receiving oral treatment experienced COVID-19 rebound

Reference	Country of Origin	Population Characteristics	Outcomes
Loza et al., 2022; Short-term Pregnancy Outcomes After Nirmatrelvir–Ritonavir Treatment for Mild-to-Moderate Coronavirus Disease 2019 (COVID-19) [47] https://doi.org/10.1097/AOG.000000000004900	USA	• Outpatient treatment of 11 pregnant women (7 received Pax)	Endpoint: No adverse perinatal effects were observed

Pax: PaxlovidTM, RR: risk ratio, RRR: relative risk ratio, IMV: invasive mechanical ventilation, ICU: intensive care unit, ER: emergency room, ARI: acute respiratory infection, ADR: adverse drug reaction, Abs: antibodies.

The effect was demonstrated even among vaccinated individuals [41] in a propensity score-matched cohort of 2260 vaccinated individuals. A primary composite outcome of all-cause ER visits, hospitalization, or death in 30 days occurred in 89 (7.87%) patients in the NMV/r cohort compared to 163 (14.4%) patients in the non-NMV/r cohort consistent with 45% relative risk reduction. In terms of all-cause emergency room visits, all-cause hospitalization, and 30-day mortality, the relative risk reduction was 41, 60, and 100%, respectively [41]. The effect was also demonstrated in high-risk patients, while Omicron was the prevailing variant. In a larger retrospective cohort study of COVID-19 patients in Israel, including adults, who were at high risk for severe COVID-19, 180,351 patients were included, of whom 135,482 (75.1%) had adequate COVID-19 vaccination status [36]. NMV/r was independently associated with a significantly decreased risk for the composite outcome of severe COVID-19 or death (HR: 0.54 (95% CI: 0.39-0.75)), while adequate COVID-19 vaccination status was associated with significantly decreased risk for the composite outcome (HR: 0.20 (95% CI: 0.17–0.22)). The magnitude of NMV/r effectiveness appears to be non-related to COVID-19 vaccination status (p-value for interaction = 0.129), even though, it exerted one of its maximum effects in the immunosuppressed subgroup of patients [36]. Similarly, among 307 patients with systemic autoimmune rheumatic disease receiving immunomodulatory regimens, only 1.3% was recorded to suffer the composite outcome of severe COVID-19 requiring hospitalization or death [48].

Even though, the magnitude of risk reduction was more prominent in the EPIC-HR trial (88%) than in studies depicting real-world experience, the lower risk reduction observed in the latter could be attributed to a number of factors including heterogenicity between the studies, such as those related to the virus variant, study design, and settings. In particular, during the EPIC-HR trial, the B.1.617.2 (Delta) strain was the circulating variant (98%), while recent real-world studies were conducted at the beginning of the fifth COVID-19 wave when predominant circulation (approximately 95%) of the BA.1 variant Omicron, which is associated with lower rates of severe cases compared to Delta, was occurring [36]. As for study design differences, the EPIC-HR trial included symptomatic adults prone to more severe disease. On the contrary, real-world evidence included patients with laboratory confirmation of SARS-CoV-2 infection [36], since no information on COVID-19-related symptoms was available. In this context NMV/r administration may have been delayed in real-world cohorts, as treatment in EPIC-HR was initiated within 5 days from symptoms' onset, while patients were included up to 5 days from SARS-CoV-2 laboratory confirmation in real life. Taking into account viral kinetics and disease pathophysiology, symptoms preceding laboratory confirmation, effectiveness is expected to be higher the earlier NMV/r is administrated. Lastly, adherence and follow up issues that have an impact on efficacy can be the least predicted and controlled for in a real-world setting.

9. Rebound Phenomena

Case reports have described SARS-CoV-2 viral rebound, meaning the recurrence of COVID-19 symptoms in some patients who have completed treatment with NMV/r. Case reports and results from the EPIC-HR trial even describe instances of increased SARS-CoV-2 RNA levels following the completion of the treatment course. In a cohort of 483 high-risk patients treated with NMV/r for COVID-19, two patients (0.4%) required hospitalization by day 30. Four patients (0.8%) experienced a rebound of symptoms, which were generally

mild, at a median of 9 days after treatment, and all resolved without additional COVID-19directed therapy [49]. In another cohort comprising data of 5287 patients aged >12 years (median age 61 years), out of whom 72.5% were vaccinated with three or more doses, 17.7% with two doses, and 7.8% were unvaccinated, only 6 (0.11%) hospitalizations and 2 deaths attributed to underlying disease and 39 (0.74%) emergency department encounters were seen in the defined 5–15-day period after NMV/r dispensation, representing <1% of NMV/r treated patients. Of note, all hospitalized patients had comorbidities or were >61 years of age [42]. Similarly, in a cohort of patients with systemic autoimmune rheumatic diseases receiving immunomodulatory regimens, rebound phenomena reached 8%; however no patient eventually presented severe disease, adding up in reassuring evidence [48].

The frequency, mechanism, and clin"cal 'mplications of these rebound events are unknown. One explanation for this rebound phenomenon is the resumption of SARS-CoV-2 viral replication following the completion of therapy, triggering a secondary immunemediated response that manifests as a recurrence of clinical symptoms. Clinical rebound corresponds to the development of a robust antibody and T-cell immune response, arguing against a high risk of disease progression, even though the presence of infectious virus supports the need for persistent isolation [50]. It is unclear if this also represents the resumption of viral replication in persons with incompletely controlled infection due to inadequate length of therapy (5 days) or a natural biphasic pattern of viral replication. Either way, rebound phenomena are predominantly self-limiting and mild in nature, rarely requiring further management.

There are currently no data on the efficacy of administering longer courses or a second course of NMV/r. Severely immunocompromised patients can experience prolonged periods of SARS-CoV-2 replication, which may lead to rapid viral evolution, especially in the immunocompromised population [51]. There are theoretical concerns that using a single antiviral agent in these patients may produce antiviral-resistant viruses. Reportedly, amounting Mpro mutations may confer resistance to NMV/r in vitro, while resistance genes have in fact been identified in community isolates [52]. However, preliminary analysis of 731 matched day 1 and day 5 samples with available sequencing data suggests no significant associations between Mpro mutations and treatment failure [27]. Additional studies are needed to assess this risk. The role of combination antiviral therapy or a longer treatment duration in treating severely immunocompromised patients is not yet known.

10. Expert Option and Future Directions

NMV/r has been a game changer in the management of COVID-19. Prompt administration prevents severe disease and poor outcomes in patients with severe co-morbidities including immunocompromise. These patients will most probably first be detected by physicians primarily attending them for other chronic diseases, and not necessarily COVID-19. It is thus pivotal that, subspecialty doctors are aware and trained to recommend and prescribe early oral treatment with NMV/r according to local regulations. In this effort, potential interactions should not halter its use, since they are expected to be less than feared in number and impact. In most cases, associated regimens can be temporarily interrupted or substituted, with no or minor clinical effects. In the context of its established efficacy and safety, a 'test-and-treat' approach should also be discussed, even in patients with an absence of known risk factors or age constraints, following a broader cost-effectiveness analysis. In line with the above, increase of public awareness and equal opportunity to care-seeking is necessary for any intervention to achieve maximum benefit [53]. Questions, however, do remain as to repeated or prolonged rounds of administration in cases of rebound or re-infections especially, in the immunocompromised population, in line with similar experimental practices with other antivirals [54]. Lastly, even though, at the moment no resistance mutations have been identified, caution and vigilance is warranted to identify and report potential future cases.

We are currently waiting for the results of the recently completed randomized doubleblind EPIC-PEP evaluating the results of protease inhibition for COVID-19 in post-exposure prophylaxis in adult household contacts of infected individuals [55]. Work in progress includes EPID-SR [56] and EPIC IC [57] assessing the effect of NMV/r in standard risk and immunocompromised participants, respectively, while open-label EPIC-Peds will give us more insight into the effect of protease inhibition [58]. Data of NMV/r in pregnancy have also been emerging at the moment with favorable results in small cohorts, even though systematic testing has not and is unlikely to be easily performed in the clinical setting in this group of patients [47].

11. Limitations of This Study

This was a literature review of data regarding experience with NMV/r use in the clinical setting throughout the last two years. Even though, the literature remains limited in comparison to other pharmaceutical regimens, experience with NMV/r is rapidly expanding in the context of the existing pandemic; hence, the authors run the risk of this review being outdated by the time is published. Moreover, we included only English language papers in our search. It is possible that experiences recorded in languages other than English were missed in this review, which thus may suffer from publication and selection bias.

12. Conclusions

The use of protease inhibitors on an outpatient basis has the potential to alleviate a significant burden caused by COVID-19 in health systems worldwide. NMV/r's efficacy rates and safety profile display much promise in this respect. It is crucial to raise awareness among health professionals and patients alike regarding the availability and use of NMV/r, especially among high-risk populations.

Author Contributions: K.A. and C.G. conceived the idea; K.A. and G.S. performed the literature search; K.A. and G.S. wrote the manuscript; K.A. and C.G. critically corrected the manuscript. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Conflicts of Interest: The authors declare no conflict of interest.

References

- Parasher, A. COVID-19: Current understanding of its Pathophysiology, Clinical presentation and Treatment. *Postgrad Med. J.* 2021, 97, 312–320. [CrossRef] [PubMed]
- Estiri, H.; Strasser, Z.H.; Murphy, S.N. Individualized prediction of COVID-19 adverse outcomes with MLHO. Sci. Rep. 2021, 11, 5322. [CrossRef] [PubMed]
- Logue, J.K.; Franko, N.M.; McCulloch, D.J.; McDonald, D.; Magedson, A.; Wolf, C.R.; Chu, H.Y. Sequelae in Adults at 6 Months After COVID-19 Infection. *JAMA Netw. Open* 2021, 4, e210830. [CrossRef] [PubMed]
- 4. NIH. COVID-19 Treatment Guidelines. Available online: https://www.covid19treatmentguidelines.nih.gov/management/ clinical-management-of-adults/nonhospitalized-adults--therapeutic-management/ (accessed on 1 November 2022).
- 5. European Centre for Disease Prevention and Control 2022. High-Risk Groups for COVID-19. Available online: www.ecdc.europa. eu/en/covid-19/high-risk-groups (accessed on 5 November 2022).
- Wu, Z.; McGoogan, J.M. Characteristics of and Important Lessons From the Coronavirus Disease 2019 (COVID-19) Outbreak in China: Summary of a Report of 72314 Cases From the Chinese Center for Disease Control and Prevention. *JAMA* 2020, 323, 1239–1242. [CrossRef] [PubMed]
- Di Fusco, M.; Shea, K.M.; Lin, J.; Nguyen, J.L.; Angulo, F.J.; Benigno, M.; Malhotra, D.; Emir, B.; Sung, A.H.; Hammond, J.L.; et al. Health outcomes and economic burden of hospitalized COVID-19 patients in the United States. *J. Med. Econ.* 2021, 24, 308–317. [CrossRef]
- 8. Scudellari, M. How the pandemic might play out in 2021 and beyond. Nature 2020, 584, 22-25. [CrossRef]
- Sallam, M. COVID-19 Vaccine Hesitancy Worldwide: A Concise Systematic Review of Vaccine Acceptance Rates. *Vaccines* 2021, 9, 160. [CrossRef]

- 10. Bergwerk, M.; Gonen, T.; Lustig, Y.; Amit, S.; Lipsitch, M.; Cohen, C.; Mandelboim, M.; Levin, E.G.; Rubin, C.; Indenbaum, V.; et al. Covid-19 Breakthrough Infections in Vaccinated Health Care Workers. *N. Engl. J. Med.* **2021**, *385*, 1474–1484. [CrossRef]
- Naaber, P.; Tserel, L.; Kangro, K.; Sepp, E.; Jürjenson, V.; Adamson, A.; Haljasmägi, L.; Rumm, A.P.; Maruste, R.; Kärner, J.; et al. Dynamics of antibody response to BNT162b2 vaccine after six months: A longitudinal prospective study. *Lancet Reg. Health Eur.* 2021, 10, 100208. [CrossRef]
- Harvey, W.T.; Carabelli, A.M.; Jackson, B.; Gupta, R.K.; Thomson, E.C.; Harrison, E.M.; Ludden, C.; Reeve, R.; Rambaut, A.; Consortium, C.-G.U.; et al. SARS-CoV-2 variants, spike mutations and immune escape. *Nat. Rev. Microbiol.* 2021, 19, 409–424. [CrossRef]
- 13. Harris, R.J.; Hall, J.A.; Zaidi, A.; Andrews, N.J.; Dunbar, J.K.; Dabrera, G. Effect of Vaccination on Household Transmission of SARS-CoV-2 in England. *N. Engl. J. Med.* **2021**, *385*, 759–760. [CrossRef] [PubMed]
- Tan, D.H.S.; Chan, A.K.; Jüni, P.; Tomlinson, G.; Daneman, N.; Walmsley, S.; Muller, M.; Fowler, R.; Murthy, S.; Press, N.; et al. Post-exposure prophylaxis against SARS-CoV-2 in close contacts of confirmed COVID-19 cases (CORIPREV): Study protocol for a cluster-randomized trial. *Trials* 2021, 22, 224. [CrossRef] [PubMed]
- 15. Tanne, J.H. COVID-19: FDA authorises pharmacists to prescribe Paxlovid. BMJ 2022, 378, o1695. [CrossRef] [PubMed]
- 16. EMA. COVID-19 Treatments. Available online: https://www.ema.europa.eu/en/human-regulatory/overview/public-health-threats/coronavirus-disease-covid-19/treatments-vaccines/covid-19-treatments (accessed on 5 November 2022).
- 17. Goyal, A.; Cardozo-Ojeda, E.F.; Schiffer, J.T. Potency and timing of antiviral therapy as determinants of duration of SARS-CoV-2 shedding and intensity of inflammatory response. *Sci. Adv.* **2020**, *6*, eabc7112. [CrossRef]
- Forrest, J.I.; Rayner, C.R.; Park, J.J.H.; Mills, E.J. Early Treatment of COVID-19 Disease: A Missed Opportunity. *Infect Dis. Ther.* 2020, 9, 715–720. [CrossRef] [PubMed]
- Gonçalves, A.; Bertrand, J.; Ke, R.; Comets, E.; de Lamballerie, X.; Malvy, D.; Pizzorno, A.; Terrier, O.; Calatrava, M.R.; Mentré, F.; et al. Timing of Antiviral Treatment Initiation is Critical to Reduce SARS-CoV-2 Viral Load. *CPT Pharmacomet. Syst. Pharm.* 2020, 9, 509–514. [CrossRef] [PubMed]
- 20. Rai, B.; Shukla, A.; Dwivedi, L.K. Incubation period for COVID-19: A systematic review and meta-analysis. *Z Gesundh. Wiss.* **2022**, *30*, 2649–2656. [CrossRef]
- 21. Eastman, R.T.; Roth, J.S.; Brimacombe, K.R.; Simeonov, A.; Shen, M.; Patnaik, S.; Hall, M.D. Remdesivir: A Review of Its Discovery and Development Leading to Emergency Use Authorization for Treatment of COVID-19. *ACS Cent. Sci.* 2020, *6*, 672–683. [CrossRef] [PubMed]
- 22. Rubin, E.J.; Baden, L.R. The Potential of Intentional Drug Development. N. Engl. J. Med. 2022, 386, 1463–1464. [CrossRef]
- 23. Yang, H.; Xie, W.; Xue, X.; Yang, K.; Ma, J.; Liang, W.; Zhao, Q.; Zhou, Z.; Pei, D.; Ziebuhr, J.; et al. Design of wide-spectrum inhibitors targeting coronavirus main proteases. *PLoS Biol.* **2005**, *3*, e324. [CrossRef]
- Owen, D.R.; Allerton, C.M.N.; Anderson, A.S.; Aschenbrenner, L.; Avery, M.; Berritt, S.; Boras, B.; Cardin, R.D.; Carlo, A.; Coffman, K.J.; et al. An oral SARS-CoV-2 M(pro) inhibitor clinical candidate for the treatment of COVID-19. *Science* 2021, 374, 1586–1593. [CrossRef] [PubMed]
- Razali, R.; Asis, H.; Budiman, C. Structure-Function Characteristics of SARS-CoV-2 Proteases and Their Potential Inhibitors from Microbial Sources. *Microorganisms* 2021, 9, 2481. [CrossRef] [PubMed]
- Yang, K.S.; Leeuwon, S.Z.; Xu, S.; Liu, W.R. Evolutionary and Structural Insights about Potential SARS-CoV-2 Evasion of Nirmatrelvir. J. Med. Chem. 2022, 65, 8686–8698. [CrossRef] [PubMed]
- Hammond, J.; Leister-Tebbe, H.; Gardner, A.; Abreu, P.; Bao, W.; Wisemandle, W.; Baniecki, M.; Hendrick, V.M.; Damle, B.; Simón-Campos, A.; et al. Oral Nirmatrelvir for High-Risk, Nonhospitalized Adults with COVID-19. *N. Engl. J. Med.* 2022, 386, 1397–1408. [CrossRef]
- 28. Ahmad, B.; Batool, M.; Ain, Q.U.; Kim, M.S.; Choi, S. Exploring the Binding Mechanism of PF-07321332 SARS-CoV-2 Protease Inhibitor through Molecular Dynamics and Binding Free Energy Simulations. *Int. J. Mol. Sci.* 2021, 22, 9124. [CrossRef]
- Vangeel, L.; Chiu, W.; De Jonghe, S.; Maes, P.; Slechten, B.; Raymenants, J.; André, E.; Leyssen, P.; Neyts, J.; Jochmans, D. Remdesivir, Molnupiravir and Nirmatrelvir remain active against SARS-CoV-2 Omicron and other variants of concern. *Antivir. Res.* 2022, 198, 105252. [CrossRef]
- Cooper, C.L.; van Heeswijk, R.P.; Gallicano, K.; Cameron, D.W. A review of low-dose ritonavir in protease inhibitor combination therapy. *Clin. Infect. Dis.* 2003, *36*, 1585–1592. [CrossRef]
- PAXLOVID SPC. 2022. Available online: https://www.ema.europa.eu/en/medicines/human/EPAR/paxlovid (accessed on 3 November 2022).
- 32. Gandhi, R.T.; Malani, P.N.; Del Rio, C. COVID-19 Therapeutics for Nonhospitalized Patients. JAMA 2022, 327, 617–618. [CrossRef]
- Bhimraj, A.; Morgan, R.L.; Shumaker, A.H.; Baden, L.; Cheng, V.C.C.; Edwards, K.M.; Gallagher, J.C.; Gandhi, R.T.; Muller, W.J.; Nakamura, M.M.; et al. Infectious Diseases Society of America Guidelines on the Treatment and Management of Patients with COVID-19. *Clin. Infect. Dis.* 2022, ciac724. [CrossRef]
- Rochwerg, B.; Siemieniuk, R.A.; Agoritsas, T.; Lamontagne, F.; Askie, L.; Lytvyn, L.; Agarwal, A.; Leo, Y.-S.; Macdonald, H.; Zeng, L.; et al. A living WHO guideline on drugs for COVID-19. *BMJ* 2020, 370, m3379. [CrossRef]
- Arbel, R.; Sagy, Y.W.; Hoshen, M.; Battat, E.; Lavie, G.; Sergienko, R.; Friger, M.; Waxman, J.G.; Dagan, N.; Balicer, R.; et al. Nirmatrelvir Use and Severe Covid-19 Outcomes during the Omicron Surge. N. Engl. J. Med. 2022, 387, 790–798. [CrossRef] [PubMed]

- Najjar-Debbiny, R.; Gronich, N.; Weber, G.; Khoury, J.; Amar, M.; Stein, N.; Goldstein, L.H.; Saliba, W. Effectiveness of Paxlovid in Reducing Severe COVID-19 and Mortality in High Risk Patients. *Clin. Infect. Dis.* 2022, ciac443. [CrossRef] [PubMed]
- Wong, C.K.H.; Au, I.C.H.; Lau, K.T.K.; Lau, E.H.Y.; Cowling, B.J.; Leung, G.M. Real-world effectiveness of early molnupiravir or nirmatrelvir-ritonavir in hospitalised patients with COVID-19 without supplemental oxygen requirement on admission during Hong Kong's omicron BA.2 wave: A retrospective cohort study. *Lancet Infect. Dis.* 2022. [CrossRef]
- Wong, C.K.H.; Au, I.C.H.; Lau, K.T.K.; Lau, E.H.Y.; Cowling, B.J.; Leung, G.M. Real-world effectiveness of molnupiravir and nirmatrelvir plus ritonavir against mortality, hospitalisation, and in-hospital outcomes among community-dwelling, ambulatory patients with confirmed SARS-CoV-2 infection during the omicron wave in Hong Kong: An observational study. *Lancet* 2022, 400, 1213–1222. [CrossRef] [PubMed]
- Yip, T.C.F.; Lui, G.C.Y.; Lai, M.S.M.; Wong, V.W.S.; Tse, Y.K.; Ma, B.H.M.; Hui, E.; Leung, M.K.W.; Chan, H.L.Y.; Hui, D.S.C.; et al. Impact of the use of oral antiviral agents on the risk of hospitalization in community COVID-19 patients. *Clin. Infect. Dis.* 2022, ciac687. [CrossRef]
- 40. Dryden-Peterson, S.; Kim, A.; Kim, A.Y.; Caniglia, E.C.; Lennes, I.; Patel, R.; Gainer, L.; Dutton, L.; Donahue, E.; Gandhi, R.T.; et al. Nirmatrelvir plus ritonavir for early COVID-19 and hospitalization in a large US health system. *medRxiv* 2022. [CrossRef]
- Ganatra, S.; Dani, S.S.; Ahmad, J.; Kumar, A.; Shah, J.; Abraham, G.M.; McQuillen, D.P.; Wachter, R.M.; E Sax, P. Oral Nirmatrelvir and Ritonavir in Non-hospitalized Vaccinated Patients with Covid-19. *Clin. Infect. Dis.* 2022, ciac673. [CrossRef]
- Malden, D.E.; Hong, V.; Lewin, B.J.; Ackerson, B.K.; Lipsitch, M.; Lewnard, J.A.; Tartof, S.Y. Hospitalization and Emergency Department Encounters for COVID-19 After Paxlovid Treatment—California, December 2021-May 2022. MMWR Morb. Mortal. Wkly. Rep. 2022, 71, 830–833. [CrossRef]
- Allen, S.; Jonhson, K.; Haddock, J.; Trinkl, J.; Nariani, S.; Joyce, B. Game Changer: Paxlovid Reduces Hospitalizations and Saves Lives. Cosmos Study. 2022. Available online: https://epicresearch.org/articles/game-changer-paxlovid-reduceshospitalizations-and-saves-lives (accessed on 1 November 2022).
- Lewnard, J.A.; Malden, D.; Hong, V.; Puzniak, L.; Kim, J.S.; Shaw, S.F.; Takhar, H.; Jodar, L.; McLaughlin, J.M.; Tartof, S.Y. Effectiveness of nirmatrelvir-ritonavir against hospital admission: A matched cohort study in a large US healthcare system. *medRxiv* 2022. [CrossRef]
- Wai, A.K.-C.; Chan, C.Y.; Cheung, A.W.-L.; Wang, K.; Chan, S.C.-L.; Lee, T.T.-L.; Luk, L.Y.-F.; Yip, E.T.-F.; Ho, J.W.-K.; Tsui, O.W.-K.; et al. Association of Molnupiravir and Nirmatrelvir-Ritonavir with preventable mortality, hospital admissions and related avoidable healthcare system cost among high-risk patients with mild to moderate COVID-19. *Lancet Reg. Health West Pac.* 2022, 100602. [CrossRef]
- Gentile, I.; Scotto, R.; Moriello, N.S.; Pinchera, B.; Villari, R.; Trucillo, E.; Ametrano, L.; Fusco, L.; Castaldo, G.; Buonomo, A.R.; et al. Nirmatrelvir/Ritonavir and Molnupiravir in the Treatment of Mild/Moderate COVID-19: Results of a Real-Life Study. *Vaccines* 2022, 10, 1731. [CrossRef] [PubMed]
- Loza, A.; Farias, R.; Gavin, N.; Wagner, R.; Hammer, E.; Shields, A. Short-term Pregnancy Outcomes After Nirmatrelvir–Ritonavir Treatment for Mild-to-Moderate Coronavirus Disease 2019 (COVID-19). Obstet. Gynecol. 2022, 140, 447–449. [CrossRef] [PubMed]
- Qian, G.; Wang, X.; Patel, N.J.; Kawano, Y.; Fu, X.; Cook, C.E.; Vanni, K.M.M.; Kowalski, E.N.; Banasiak, E.P.; Banasiak, E.P.; et al. Outcomes with and without outpatient SARS-CoV-2 treatment for patients with COVID-19 and systemic autoimmune rheumatic diseases: A retrospective cohort study. *medRxiv* 2022. [CrossRef]
- Ranganath, N.; O'Horo, J.C.; Challener, D.W.; Tulledge-Scheitel, S.M.; Pike, M.L.; O'Brien, R.M.; Razonable, R.R.; Shah, A. Rebound Phenomenon after Nirmatrelvir/Ritonavir Treatment of Coronavirus Disease-2019 in High-Risk Persons. *Clin. Infect. Dis.* 2022, ciac481. [CrossRef]
- Epling, B.P.; Rocco, J.M.; Boswell, K.L.; Laidlaw, E.; Galindo, F.; Kellogg, A.; Das, S.; Roder, A.; Ghedin, E.; Kreitman, A.; et al. Clinical, Virologic, and Immunologic Evaluation of Symptomatic Coronavirus Disease 2019 Rebound Following Nirmatrelvir/Ritonavir Treatment. *Clin. Infect. Dis.* 2022, ciac663. [CrossRef]
- 51. Truong, T.T.; Ryutov, A.; Pandey, U.; Yee, R.; Goldberg, L.; Bhojwani, D.; Aguayo-Hiraldo, P.; Pinsky, B.A.; Pekosz, A.; Shen, L.; et al. Persistent SARS-CoV-2 infection and increasing viral variants in children and young adults with impaired humoral immunity. *medRxiv* 2021. [CrossRef]
- 52. Motyan, J.A.; Mahdi, M.; Hoffka, G.; Tozser, J. Potential Resistance of SARS-CoV-2 Main Protease (Mpro) against Protease Inhibitors: Lessons Learned from HIV-1 Protease. *Int. J. Mol. Sci.* **2022**, *23*, 3507. [CrossRef]
- Bajema, K.; Wang, X.Q.; Hynes, D.M.; Rowneki, M.; Hickok, A.; Cunningham, F.; Bohnert, A.; Boyko, E.J.; Iwashyna, T.J.; Maciejewski, M.L.; et al. Early Adoption of Anti–SARS-CoV-2 Pharmacotherapies Among US Veterans With Mild to Moderate COVID-19, January and February 2022. *JAMA Netw. Open* 2022, *5*, e2241434. [CrossRef]
- Helleberg, M.; Niemann, C.U.; Moestrup, K.S.; Kirk, O.; Lebech, A.M.; Lane, C.; Lundgren, J. Persistent COVID-19 in an Immunocompromised Patient Temporarily Responsive to Two Courses of Remdesivir Therapy. J. Infect. Dis. 2020, 222, 1103–1107. [CrossRef]
- 55. ClinicalTrials.gov. (2022). NCT. Available online: www.clinicaltrials.gov/ct2/show/NCT05047601 (accessed on 1 November 2022).
- 56. ClinicalTrials.gov. (2022). NCT. Available online: https://clinicaltrials.gov/ct2/show/NCT05011513 (accessed on 1 November 2022).
- 57. ClinicalTrials.gov. (2022). NCT. Available online: https://clinicaltrials.gov/ct2/show/NCT05438602 (accessed on 1 November 2022).
- 58. ClinicalTrials.gov. (2022). NCT. Available online: https://clinicaltrials.gov/ct2/show/NCT05261139 (accessed on 1 November 2022).